Author’s response to reviews

Title: First and second eye cataract surgery and driver self-regulation among older drivers with bilateral cataract: a prospective cohort study

Authors:

Seraina Agramunt (seraina.agramunt@curtin.edu.au)

Lynn Meuleners (l.meuleners@curtin.edu.au)

Michelle Fraser (M.Fraser@exchange.curtin.edu.au)

Kyle Chow (C.Chow@curtin.edu.au)

Jonathon Ng (jonathon.ng@uwa.edu.au)

Vignesh Raja (Vignesh.Raja@health.wa.gov.au)

Version: 3 Date: 02 Feb 2018

Author’s response to reviews:

Dear Editor,

Thank you for your comments about author contributions. We have added more detail to this section and all are justified for authorship.

Authors’ contributions: SA collected the data, conducted the data analysis, interpretation of data, and drafted the manuscript. LM led the design of the study, conducted the data analysis, interpretation of data, and drafted the manuscript. MF made substantial contributions to the conception and design of the study, analysed and interpreted the data and was involved in drafting and revising the manuscript. KC contributed to the design of the study, analysis of data and drafting and revising of the manuscript. JN contributed to the acquisition of data, interpretation of data and was involved in revising the manuscript critically for important intellectual content. VR made substantial contributions to the acquisition of data and the interpretation of data and revised the manuscript critically for important intellectual content. All authors read, revised and approved the final manuscript.

Kind regards,

Professor Lynn Meuleners